Author's response to reviews

Title: Duration of viral shedding in hospitalized patients infected with pandemic H1N1

Authors:

Silvia Meschi (silvia.meschi@inmi.it)
Marina Selleri (marina.selleri@inmi.it)
Eleonora Lalle (eleonora.lalle@inmi.it)
Licia Bordi (licia.bordi@inmi.it)
Maria B Valli (mariabeatrice.valli@inmi.it)
Federica Ferraro (federica.ferraro@inmi.it)
Giuseppe Ippolito (giuseppe.ippolito@inmi.it)
Nicola Petrosillo (nicola.petrosillo@inmi.it)
Francesco N Lauria (francesco.lauria@inmi.it)
Maria R Capobianchi (maria.capobianchi@inmi.it)

Version: 9 Date: 18 May 2011

Author's response to reviews: see over
Dear Editor, we are resubmitting our paper entitled:

**RE: manuscript: MS: 1195715970434904**

**Title:** DURATION OF VIRAL SHEDDING IN HOSPITALIZED PATIENTS INFECTED WITH PANDEMIC H1N1

We have noticed a typo in the previous version, that has been corrected (page 10). Furthermore we have performed other small typographical changes, that have not been highlighted in the revised version.

Please, include in future communication, a request of receipt notification, as during the processing of our paper we have experienced a lot of problems in receiving the e-mail messages from the Journal.

Best regards

The corresponding author

Dr. Maria R. Capobianchi
The National Institute for Infectious Diseases “L. Spallanzani”
Via Portuense 292 00149 Rome, Italy
Tel +39-0655170434 Fax +39-065594555
e-mail: maria.capobianchi@inmi.it